Author's response to reviews

Title: Final 2 year results of the Vascular Imaging of acute Stroke for Identifying predictors of clinical Outcome and recurrent ischemic eveNts (VISION) study.

Authors:

Shelagh B Coutts (scoutts@ucalgary.ca)
Michael D Hill (hillmd@ucalgary.ca)
Misha Eliasziw (eliasziw@ucalgary.ca)
Karyn Fischer (Karyn.Fischer@CalgaryHealthRegion.ca)
Andrew M Demchuk (ademchuk@ucalgary.ca)

Version: 4 Date: 12 April 2011

Author's response to reviews: see over
We have made the changes to raw and adjusted P values in Table 1 and Table 2 as suggested by reviewer 3.